Author's response to reviews

Title: Psycho-education with problem solving (PEPS) therapy for adults with personality disorder: A pragmatic multi-site community-based randomised clinical trial

Authors:

Mary Mary McMurrnan (Mary.McMurrnan@nottingham.ac.uk)
Mike J Crawford (M.Crawford@imperial.ac.uk)
Joseph G Reilly (J.G.Reilly@durham.ac.uk)
Paul McCrone (Paul.Mccrone@iop.kcl.ac.uk)
Hywel Williams (Hywel.Williams@nottingham.ac.uk)
Clive E Adams (Clive.Adams@nottingham.ac.uk)
Paul Moran (Paul.Moran@iop.kcl.ac.uk)
Conor Duggan (Conor.Duggan@nottingham.ac.uk)
Juan Delport (Juan.Delport@wales.nhs.uk)
Diane Whitham (Diane.Whitham@nottingham.ac.uk)
Florence Day (Florence.Day@nottingham.ac.uk)

Version: 5 Date: 17 August 2011

Author's response to reviews:

I have checked for Typos. Figure 2 is now on a single page.